Author's response to reviews

Title: Gastrointestinal cyst attached to the gallbladder: a case report

Authors:

Angel I Popkharitov (popkharitov@abv.bg)
Maya V Gulubova (mgulubova@hotmail.com)
Angel D Dandov (angel_dandov@mail.bg)
Dimitar P Sivrev (dsivrev@yahoo.com)

Version: 5 Date: 31 December 2007

Author's response to reviews:

MEDICAL FACULTY, THRACIAN UNIVERSITY,
Department of General and Clinical Pathology
Tel: (+359) 42 4 00 96, Fax: (+359) 42 600 705
e.mail: popkharitov@abv.bg

31.12.07, Stara Zagora

To: The Chief Editor of
The Journal of Medical Case Reports

Dear Sirs,

Enclosed, please, find our revised manuscript entitled “Gastrointestinal cyst attached to the gallbladder: a case report.”

According to your recommendation last version of our manuscript was reviewed by a native English speaker.

We hope that the information and the results of our work will be of interest to the readers of Journal of Medical Case Reports and will be accepted for publication in the journal.

Have a happy and a very special New Year!

Sincerely Yours,

Dr Angel Popkharitov, MD, PhD
Dr Dimitar Sivrev, MD, PhD
Thracian University, Medical Faculty Thracian University, Medical Faculty
Dept. of Surgery Dept. of Anatomy
11 Armeiska Str., 11 Armeiska Str
6000 Stara Zagora, Bulgaria 6000 Stara Zagora, Bulgaria
tel.: ++ 359 42 644 450
fax: ++359 42 600 710
popkharitov@abv.bg dsivrev@yahoo.com